Author's response to reviews

Title: Second trimester amniotic fluid cytokine concentrations, Ureaplasma spp. colonisation status and sexual activity as predictors of preterm birth in Chinese and Australian women

Authors:

Matthew S Payne (matthew.payne@uwa.edu.au)
Zhenhua Feng (fengzhenhua1222@163.com)
Shaofu Li (shaofu.li@uwa.edu.au)
Dorota A Doherty (dorota.doherty@uwa.edu.au)
Biyun Xu (biyunxu@163.com)
Jie Li (jie1967@126.com)
Lenan Liu (victorialiu1203@sina.com)
Jeffrey A Keelan (jeff.keelan@uwa.edu.au)
Yi-Hua Zhou (zgr03summer@126.com)
Jan E Dickinson (jan.dickinson@uwa.edu.au)
Yali Hu (yalihu@nju.edu.cn)
John P Newnham (john.newnham@uwa.edu.au)

Version: 2
Date: 6 June 2014

Author's response to reviews:

Dear Dr Trinidad,

Please find below a list of the changes you requested in your email dated 6 June 2014, which we have addressed as follows:

- email addresses of all authors have been included in the Title Page
- Line and page numbers have been inserted in the text of the manuscript
- a "competing Interests" section has been added between the Conclusions and Author's Contributions
- Author's Contributions section is now added

Many thanks for your time and assistance,

Best wishes,

John Newnham